\begin{thebibliography}{10}

\bibitem{adams1986}
Edward~N. Adams~III.
\newblock N-trees as nestings: Complexity, similarity, and consensus.
\newblock {\em J. Classif.}, 3:299--317, 1986.

\bibitem{aho1981}
Alfred~V. Aho, Yehoshua Sagiv, Thomas~G. Szymanski, and Jeffrey~D. Ullman.
\newblock Inferring a tree from lowest common ancestors with an application to
  the optimization of relational expressions.
\newblock {\em SIAM J. Comput.}, 10:405--421, 1981.

\bibitem{bandelt1999}
Hans-J. Bandelt, Peter Forster, and Arne R{\"o}hl.
\newblock Median-joining networks for inferring intraspecific phylogenies.
\newblock {\em Mol. Biol. Evol.}, 16:37--48, 1999.

\bibitem{bandelt1995}
Hans-J. Bandelt, Peter Forster, Bryan~C. Sykes, and Martin~B. Richards.
\newblock Mitochondrial portraits of human populations using median networks.
\newblock {\em Genetics}, 141:743--753, 1995.

\bibitem{baroni2004}
Mihaela Baroni, Charles Semple, and Mike Steel.
\newblock A framework for representing reticulate evolution.
\newblock {\em Ann. Combin.}, 8:391--408, 2004.

\bibitem{baum1992}
Bernard~R. Baum.
\newblock Combining trees as a way of combining data sets for phylogenetic
  inference, and the desirability of combining gene trees.
\newblock {\em Taxon}, 41:3--10, 1992.

\bibitem{bendor1998}
Amir Ben-dor, Benny Chor, Dan Graur, Ophir Ron, and Dan Pelleg.
\newblock Constructing phylogenies from quartets: Elucidation of eutherian
  superordinal relationships.
\newblock {\em J. Comput. Biol.}, 5:377--390, 1998.

\bibitem{bininda2002}
Olaf R.~P. Bininda-Emonds.
\newblock {MRP} supertree construction in the consensus setting.
\newblock In M.~F. Janowitz, F.-J. Lapointe, F.~R. McMorris, B.~Mirkin, and
  F.~S. Roberts, editors, {\em DIMACS Series in Discrete Mathematics and
  Theoretical Computer Science}, volume~61, pages 231--242. American
  Mathematical Society, Providence, Rhode Island, 2003.

\bibitem{brauer2002}
Matthew~J. Brauer, Mark~T. Holder, Laurie~A. Dries, Derrick~J. Zwickl, Paul~O.
  Lewis, and David~M. Hillis.
\newblock Genetic algorithms and parallel processing in maximum-likelihood
  phylogeny inference.
\newblock {\em Mol. Biol. Evol.}, 19:1717--1726, 2002.

\bibitem{bremer1990}
K.~Bremer.
\newblock Combinable component consensus.
\newblock {\em Cladistics}, 6:369--372, 1990.

\bibitem{bryant2004}
David Bryant and Vincent Moulton.
\newblock Neighbor-net: An agglomerative method for the construction of
  phylogenetic networks.
\newblock {\em Mol. Biol. Evol.}, 21:255--265, 2004.

\bibitem{bryant1995}
David Bryant and Mike Steel.
\newblock Extension operations on sets of leaf-labeled trees.
\newblock {\em Adv. Appl. Math.}, 16:425--453, 1995.

\bibitem{chen2003}
D.~Chen, L.~Diao, O.~Eulenstein, D.~Fern\'{a}ndez-Baca, and M.~J. Sanderson.
\newblock Flipping: A supertree construction method.
\newblock In M.~F. Janowitz, F.-J. Lapointe, F.~R. McMorris, B.~Mirkin, and
  F.~S. Roberts, editors, {\em DIMACS Series in Discrete Mathematics and
  Theoretical Computer Science}, volume~61, pages 135--160. American
  Mathematical Society, Providence, Rhode Island, 2003.

\bibitem{cooke1980}
Peter Cooke.
\newblock Optimal linear estimation of bounds of random variables.
\newblock {\em Biometrika}, 67:257--258, 1980.

\bibitem{cormen2001}
Thomas~H. Cormen, Charles~E. Leiserson, Ronald~L. Rivest, and Clifford Stein.
\newblock {\em Introduction to Algorithms}.
\newblock The MIT Press, Cambridge, Massachusetts, 2 edition, 2001.

\bibitem{dagum1998}
Leonardo Dagum and Ramesh Menon.
\newblock {OpenMP}: An industry-standard {API} for shared-memory programming.
\newblock {\em IEEE Comput. Sci. Eng.}, 5:46--55, 1998.

\bibitem{queiroz1995}
Alan de~Queiroz, Michael~J. Donoghue, and Junhyong Kim.
\newblock Separate versus combined analysis of phylogenetic evidence.
\newblock {\em Annu. Rev. Ecol. Syst.}, 26:657--681, 1995.

\bibitem{dress1996}
Andreas Dress, Daniel Huson, and Vincent Moulton.
\newblock Analyzing and visualizing sequence and distance data using
  \textsc{SplitsTree}.
\newblock {\em Discr. Appl. Math.}, 71:95--109, 1996.

\bibitem{dress2004}
Andreas W.~M. Dress and Daniel~H. Huson.
\newblock Constructing splits graphs.
\newblock {\em IEEE/ACM Trans. Comput. Biol. Bioinform.}, 1:109--115, 2004.

\bibitem{felsenstein1978}
Joseph Felsenstein.
\newblock The number of evolutionary trees.
\newblock {\em Syst. Zool.}, 27:27--33, 1978.

\bibitem{fisher1912}
R.~A. Fisher.
\newblock On an absolute criterion for fitting frequency curves.
\newblock {\em Messenger in Mathematics}, 41:155--160, 1912.

\bibitem{gropp1998}
William Gropp, Steven Huss-Lederman, Andrew Lumsdaine, Ewing Lusk, Bill
  Nitzberg, William Saphir, and Marc Snir.
\newblock {\em {MPI}: The Complete Reference - The {MPI} Extensions}, volume~2.
\newblock The MIT Press, Cambridge, Massachusetts, 2 edition, 1998.

\bibitem{guindon2003}
St\'{e}phane Guindon and Olivier Gascuel.
\newblock A simple, fast, and accurate algorithm to estimate large phylogenies
  by maximum likelihood.
\newblock {\em Syst. Biol.}, 52:696--704, 2003.

\bibitem{hendy1989}
Michael~D. Hendy and David Penny.
\newblock A framework for the quantitative study of evolutionary trees.
\newblock {\em Syst. Zool.}, 38:297--309, 1989.

\bibitem{holland2004}
Barbara~R. Holland, Katharina~T. Huber, Vincent Moulton, and Peter~J. Lockhart.
\newblock Using consensus networks to visualize contradictory evidence for
  species phylogeny.
\newblock {\em Mol. Biol. Evol.}, 21:1459--1461, 2004.

\bibitem{huber2002}
Katharina~T. Huber, Michael Langton, David Penny, Vincent Moulton, and Michael
  Hendy.
\newblock Spectronet: a package for computing spectra and median networks.
\newblock {\em Appl. Bioinform.}, 20:159--161, 2002.

\bibitem{huson1997}
Daniel~H. Huson.
\newblock {SplitsTree}: A program for analyzing and visualizing evolutionary
  data.
\newblock Materialien/Preprints 110, Universit\"at Bielefeld,
  Forschungsschwerpunkt Mathematisierung -- Stukturbildungsprozesse, Bielefeld,
  1997.

\bibitem{huson1998}
Daniel~H. Huson.
\newblock {SplitsTree}: Analyzing and visualizing evolutionary data.
\newblock {\em Bioinformatics}, 14:68--73, 1998.

\bibitem{huson2004a}
Daniel~H. Huson, Tobias Dezulian, Tobias Kl{\"o}pper, and Mike~A. Steel.
\newblock Phylogenetic super-networks from partial trees.
\newblock {\em IEEE/ACM Trans. Comput. Biol. Bioinform.}, 1:151--158, 2004.

\bibitem{jansson2004}
Jesper Jansson and Wing-Kin Sung.
\newblock The maximum agreement of two nested phylogenetic networks.
\newblock In {\em Proceedings of the 15th International Symposium on Algorithms
  and Computation (ISAAC 2004)}, volume 3341 of {\em Lecture Notes in Computer
  Science}, pages 581--593, New York, 2004. Springer.

\bibitem{keane2005}
T.~M. Keane, T.~J. Naughton, S.~A.~A. Travers, J.~O. McInerney, and G.~P.
  McCormack.
\newblock {DPRml}: distributed phylogeny reconstruction by maximum likelihood.
\newblock {\em Bioinformatics}, 21:969--974, 2005.

\bibitem{lemmon2002}
Alan~R. Lemmon and Michel~C. Milinkovitch.
\newblock The metapopulation genetic algorithm: An efficient solution for the
  problem of large phylogeny estimation.
\newblock {\em Proc. Natl. Acad. Sci. USA}, 99:10516--10521, 2002.

\bibitem{maddison1992a}
David~R. Maddison, Maryellen Ruvolo, and David~L. Swofford.
\newblock Geographic origins of human mitochondrial {DNA}: phylogenetic
  evidence from control region sequences.
\newblock {\em Syst. Biol.}, 41:111--124, 1992.

\bibitem{maddison1984}
Wayne~P. Maddison, Michael~J. Donoghue, and David~R. Maddison.
\newblock Outgroup analysis and parsimony.
\newblock {\em Syst. Zool.}, 33:83--103, 2005.

\bibitem{makarenkov2004}
Vladimir Makarenko and Pierre Legendre.
\newblock From a phylogenetic tree to a reticulated network.
\newblock {\em J. Comput. Biol.}, 11:195--212, 2004.

\bibitem{margush1981}
Tim Margush and Fred~R. McMorris.
\newblock Consensus n-trees.
\newblock {\em Bull. Math. Biol.}, 43:239--244, 1981.

\bibitem{martin2004}
William Martin and T.~Martin Embley.
\newblock Evolutionary biology: Early evolution comes full circle.
\newblock {\em Nature}, 431:134--137, 2004.

\bibitem{mcmorris1983}
Fred~R. McMorris and Dean Neumann.
\newblock Consensus functions defined on trees.
\newblock {\em Math. Soc. Sci}, 4:131--136, 1983.

\bibitem{moret2004}
Bernard~M.E. Moret, Luay Nakhleh, Tandy Warnow, C.~Randal Linder, Anna Tholse,
  Anneke Padolina, Jerry Sun, and Ruth Timme.
\newblock Phylogenetic networks: Modeling, reconstructibility, and accuracy.
\newblock {\em IEEE/ACM Trans. Comput. Biol. Bioinform.}, 1:13--23, 2004.

\bibitem{nakhleh2004}
Luay Nakhleh, Allen Clement, Tandy Warnow, C.~Randal Linder, and Bernard M.~E.
  Moret.
\newblock Quality measures for phylogenetic networks.
\newblock UNM Computer Science Tech-Reports TR-CS-2004-06, University New
  Mexico, Albuquerque, NM, USA, January 2004.

\bibitem{nixon1993}
Kevin~C. Nixon and James~M. Carpenter.
\newblock On outgroups.
\newblock {\em Cladistics}, 9:413--426, 1993.

\bibitem{olsen1994}
Gary~J. Olsen, Hideo Matsuda, Ray Hagstrom, and Ross Overbeek.
\newblock {fastDNAml}: A tool for construction of phylogenetic trees of {DNA}
  sequences using maximum likelihood.
\newblock {\em Comput. Appl. Biosci.}, 10:41--48, 1994.

\bibitem{page2002d}
Roderic D.~M. Page.
\newblock Modified mincut supertrees.
\newblock In {\em Proceedings of the 2nd Workshop on Algorithms in
  Bioinformatics (WABI 2002)}, volume 2452 of {\em Lecture Notes in Computer
  Science}, pages 537--551, New York, 2002. Springer.

\bibitem{page1998}
Roderic D.~M. Page and Edward~C. Holmes.
\newblock {\em Molecular Evolution: A Phylogenetic Approach}.
\newblock Blackwell Science, Oxford, 1998.

\bibitem{purvis1995}
Andy Purvis.
\newblock A composite estimate of primate phylogeny.
\newblock {\em Philos. Trans. R. Soc. Lond. Ser. B}, 348:405--421, 1995.

\bibitem{ragan1992}
M.~A. Ragan.
\newblock Phylogenetic inference based on matrix representation of trees.
\newblock {\em Mol. Phylogenet. Evol.}, 1:53--58, 1992.

\bibitem{roberts2003}
David~L. Roberts and Andrew~R. Solow.
\newblock Flightless birds: When did the dodo become extinct?
\newblock {\em Nature}, 426:245, 2003.

\bibitem{robinson2001}
Marc Robinson-Rechavi and Dan Graur.
\newblock Usage optimization of unevenly sampled data through the combination
  of quartet trees: An eutherian draft phylogeny based on 640 nuclear and
  mitochondrial proteins.
\newblock {\em Isr. J. Zool.}, 47:259--270, 2001.

\bibitem{salamin2002}
Nicolas Salamin, Trevor~R. Hodkinson, and Vincent Savolainen.
\newblock Building supertrees: An empirical assessment using the grass family
  (poaceae).
\newblock {\em Syst. Biol.}, 51:136--150, 2002.

\bibitem{sanderson1998}
Michael~J. Sanderson, Andy Purvis, and Chris Henze.
\newblock Phylogenetic supertrees: Assembling the trees of life.
\newblock {\em TREE}, 13:105--109, 1998.

\bibitem{sanderson2002a}
Michael~J. Sanderson and H.~Bradley Shaffer.
\newblock Troubleshooting molecular phylogenetic analyses.
\newblock {\em Annu. Rev. Ecol. Syst.}, 33:49--72, 2002.

\bibitem{schmidt2003a}
Heiko~A. Schmidt, Ekkehard Petzold, Martin Vingron, and Arndt von Haeseler.
\newblock Molecular phylogenetics: Parallelized parameter estimation and
  quartet puzzling.
\newblock {\em J. Parallel Distrib. Comput.}, 63:719--727, 2003.

\bibitem{schmidt2002}
Heiko~A. Schmidt, Korbinian Strimmer, Martin Vingron, and Arndt von Haeseler.
\newblock {TREE-PUZZLE}: Maximum likelihood phylogenetic analysis using
  quartets and parallel computing.
\newblock {\em Bioinformatics}, 18:502--504, 2002.

\bibitem{semple2000}
Charles Semple and Mike Steel.
\newblock A supertree method for rooted trees.
\newblock {\em Discr. Appl. Math.}, 105:147--158, 2000.

\bibitem{smith1994}
Andrew~B. Smith.
\newblock Rooting molecular trees: problems and strategies.
\newblock {\em Biol. J. Linn. Soc.}, 51:279--292, 1994.

\bibitem{snir1998}
Marc Snir, Steve~W. Otto, Steven Huss-Lederman, David~W. Walker, and Jack
  Dongarra.
\newblock {\em {MPI}: The Complete Reference - The {MPI} Core}, volume~1.
\newblock The MIT Press, Cambridge, Massachusetts, 2 edition, 1998.

\bibitem{stamatakis2005}
Alexandros~P. Stamatakis, Thomas Ludwig, and Harald Meier.
\newblock {RAxML-III}: a fast program for maximum likelihood-based inference of
  large phylogenetic trees.
\newblock {\em Bioinformatics}, 21:456--463, 2005.

\bibitem{stewart2001}
Craig~A. Stewart, David Hart, Donald~K. Berry, Gary~J. Olsen, Eric~A. Wernert,
  and William Fischer.
\newblock Parallel implementation and performance of {fastDNAml} - a program
  for maximum likelihood phylogenetic inference.
\newblock In {\em Proceedings of the International Conference on High
  Performance Computing and Communications - SC2001}, pages 191--201, November
  2001.

\bibitem{strimmer2000}
Korbinian Strimmer and Vincent Moulton.
\newblock Likelihood analysis of phylogenetic networks using directed graphical
  models.
\newblock {\em Mol. Biol. Evol.}, 17:875--881, 2000.

\bibitem{strimmer2001}
Korbinian Strimmer, Carsten Wiuf, and Vincent Moulton.
\newblock Recombination analysis using directed graphical models.
\newblock {\em Mol. Biol. Evol.}, 18:97--99, 2001.

\bibitem{swofford1996}
David~L. Swofford, Gary~J. Olsen, Peter~J. Waddell, and David~M. Hillis.
\newblock Phylogeny reconstruction.
\newblock In David~M. Hillis, Craig Moritz, and Barbara~K. Mable, editors, {\em
  Molecular Systematics}, pages 407--514. Sinauer Associates, Sunderland,
  Massachusetts, 2 edition, 1996.

\bibitem{tarrio2001}
Rosa Tarr{\'i}o, Francisco Rodr{\'i}guez-Trelles, and Francisco~J. Ayala.
\newblock Shared nucleotide composition biases among species and their impact
  on phylogenetic reconstructions of the drosophilidae.
\newblock {\em Mol. Biol. Evol.}, 18:1464--1473, 2001.

\bibitem{vinh2005}
Le~Sy Vinh, Heiko~A. Schmidt, and Arndt von Haeseler.
\newblock {PhyNav}: A novel approach to reconstruct large phylogenies.
\newblock In Claus Weihs and Wolfgang Gaul, editors, {\em Classification - The
  Ubiquitous Challenge}, Studies in Classification, Data Analysis, and
  Knowledge Organization, pages 386--393. Springer, Heidelberg, 2005.

\bibitem{vinh2004}
Le~Sy Vinh and Arndt von Haeseler.
\newblock {IQPNNI}: Moving fast through tree space and stopping in time.
\newblock {\em Mol. Biol. Evol.}, 21:1565--1571, 2004.

\bibitem{haeseler1993}
Arndt von Haeseler and Gary~A. Churchill.
\newblock Network models for sequence evolution.
\newblock {\em J. Mol. Evol.}, 37:77--85, 1993.

\bibitem{wang2001}
Lusheng Wang, Kaizhong Zhang, and Louxin Zhang.
\newblock Perfect phylogenetic networks with recombination.
\newblock {\em J. Comput. Biol.}, 8:69--78, 2001.

\bibitem{wheeler1990}
Ward~C. Wheeler.
\newblock Nucleic acid sequence phylogeny and random outgroups.
\newblock {\em Cladistics}, 6:363--368, 2005.

\bibitem{wolf2000}
Marty~J. Wolf, Simon Easteal, Margaret Kahn, Brendan~D. McKay, and Lars~S.
  Jermiin.
\newblock {TrExML}: a maximum-likelihood approach for extensive tree-space
  exploration.
\newblock {\em Bioinformatics}, 16:383--394, 2000.

\bibitem{zhou2004a}
B.~B. Zhou, M.~Till, A.~Y. Zomaya, and L.~S. Jermiin.
\newblock Parallel implementation of maximum likelihood methods for
  phylogenetic analysis.
\newblock In {\em Proceedings of the 18th International Parallel and
  Distributed Processing Symposium (IPDPS2004)}, pages 237a 1--6, Los Alamitos,
  USA, 2004. IEEE.

\bibitem{zillig1991}
Wolfgang Zillig.
\newblock Comparative biochemistry of {Archaea} and {Bacteria}.
\newblock {\em Curr. Opin. Genet. Dev.}, 1:544--551, 1991.

\end{thebibliography}

\newcommand{\mcE}{\mathcal{E}}                  % cal E set
\newcommand{\mcF}{\mathcal{F}}                  % cal F set
\newcommand{\mcL}{\mathcal{L}}                  % cal L set
\newcommand{\mcQ}{\mathcal{Q}}                  % cal Q set
\newcommand{\mcV}{\mathcal{V}}                  % cal V set
\newcommand{\mcS}{\mathcal{S}}                  % cal S set
\newcommand{\mcT}{\mathcal{T}}                  % cal T set

\newcommand{\mbbA}{\mathbb{A}}                  % bf S matr


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%5




\section{Introduction}
The large amounts of molecular sequence data currently available
serve two needs in the phylogenetic analysis of the
relationship of species. On the one hand, the number
of interesting species available for analysis grows (vertical growth). 
On the other hand, more different sequences per species get available 
(horizontal growth).
Unfortunately, the number of sequences is not distributed evenly 
among species of interest. This often makes collecting a dataset 
for phylogenetic analysis a painful decision on the trade-off 
between the amount of taxa and the number of sequences.

While methods abound to infer phylogenies from a set of 
aligned sequences \cite{swofford1996}, only a small number of methods 
exists to combine data of different genes, proteins, or genomic areas for 
joint analysis. 

There has been a large ongoing debate how to combine different 
datasets to reconstruct trees (cf. \cite{queiroz1995},
\cite{bininda2002}, and \cite{page1998}, chap.~8 for review). 
Two paradigms have been discussed, which we will classify by
the 'distance' of the combination event from the underlying data
into \emph{low level} and \emph{high level} methods
(cf. Fig.~\ref{fig:combination-level}). 
Note that \emph{low} and \emph{high} does not imply a quality rating.

The first paradigm is \emph{total evidence} (also often called 
\emph{combined} or \emph{simultaneous analysis}) where the data 
is combined directly by concatenating the alignments. 
Hence, \emph{total evidence} methods will 
be referred to as \emph{low level methods} 
(cf. Fig.~\ref{fig:combination-level}).

The other paradigm is the so-called \emph{separate analysis}
(also called \emph{taxonomic congruence} and \emph{consensus}
or \emph{supertree approach}). Such methods combine trees reconstructed
separately from the single datasets. Since the combination takes place 
far from the underlying data, such methods are referred to as 
\emph{high level methods} (cf. Fig.~\ref{fig:combination-level}).

Both paradigms have their advantages and drawbacks (for review see 
\cite{queiroz1995}, \cite{bininda2002}, and references therein). 
Major criticisms are, e.g., the problems to jointly model the evolutionary 
process for the concatenated dataset in \emph{low level methods} and
the loss of information in the \emph{high level methods} since the 
underlying data is in general not considered when the trees are
combined. These arguments gave rise to the design of an alternative method 
which will be proposed in this chapter. 

The method presented here follows a \emph{medium level} paradigm, 
according to the level of combination (cf. Fig.~\ref{fig:combination-level}).
Before particularizing the procedure (section \ref{sec:quartet-combin}), 
commonly used \emph{high} and \emph{low level methods} are described
(section \ref{sec:supertree-methods}). 
Then the new method is applied to a biological dataset in section 
\ref{sec:grass-tree}. The results are discussed in the context of 
\emph{total evidence} and \emph{consensus/supertree methods}. 
Finally, problems will be discussed and an outlook on further 
improvement and extensions will be given.

\section{Methods to Combine Datasets}
\label{sec:supertree-methods}
\begin{figure}[t!]

   \centering
   \resizebox{\textwidth}{!}{
   %\begin{tabular}{|c|p{.20\textwidth}|c|}\hline
   \begin{tabular}{|c|p{.18\textwidth}|p{.75\textwidth}|}\hline
	combination & \parbox{.18\textwidth}{method}    & 
					\parbox{.75\textwidth}{~\\
					\includegraphics[width=.75\textwidth]%
					{fig-missing/combination-level}\\}\\ 
					\hline
					\hline
	low level & \parbox{.18\textwidth}{\emph{Total Evidence}}   & 
					\parbox{.75\textwidth}{~\\
					\includegraphics[width=.75\textwidth]%
					{fig-missing/combination-total}\\}\\
					 \hline
	high level & \parbox{.18\textwidth}{\emph{Consensus}}       & 
					\parbox{.75\textwidth}{~\\
					\includegraphics[width=.75\textwidth]%
					{fig-missing/combination-consensus}\\}\\ 
					%\hline
	high level & \parbox{.18\textwidth}{\emph{MRP}, \emph{MRF}} & 
					\parbox{.75\textwidth}{~\\
					\includegraphics[width=.75\textwidth]%
					{fig-missing/combination-mrp}\\}\\ 
					%\hline
	high level & \parbox{.18\textwidth}{\sloppy\textsc{ModMinCut}, \textsc{MinCut\\Supertree}} 
                                      & \parbox{.75\textwidth}{~\\
					\includegraphics[width=.75\textwidth]%
					{fig-missing/combination-mincut}\\}\\ 
					%\hline
	high level & \parbox{.18\textwidth}{\sloppy\emph{Quartet-based supertree}} 
	                              & \parbox{.75\textwidth}{~\\
					\includegraphics[width=.75\textwidth]%
					{fig-missing/combination-quartet}\\}\\ 
					\hline
	medium level & \parbox{.18\textwidth}{\sloppy\emph{Medium level combination}}        
	                              & \parbox{.75\textwidth}{~\\
					\includegraphics[width=.75\textwidth]%
					{fig-missing/combination-puzzle}\\}\\ 
					\hline
   \end{tabular}}
   \caption[Levels of Dataset Combination]
           {Level of combination with regard to distance from the underlying 
            datasets}
   \label{fig:combination-level}
\end{figure}

%%%%%%%%%%%%
\subsection{\emph{Low Level} Combination: Total Evidence}

As mentioned above, \emph{low level methods} are commonly referred to as 
\emph{total evidence}, \emph{combined} or \emph{simultaneous analysis} 
methods.
Claiming that all information (evidence) available should be used 
for phylogenetic analysis, all single datasets are combined into 
one single \emph{'supermatix'} \cite{sanderson1998} as shown in 
Fig.~\ref{fig:combination-level}. This is achieved 
by concatenating all source alignments filling missing data with
gap characters. The overall alignment is then used as input for
phylogenetic analysis.
This method seems unproblematic for complete datasets, where
for each species one sequence in each source dataset exists.
How \emph{total evidence} approaches handle missing data
crucially depends on the method used to reconstruct a phylogeny from the 
concatenated alignment.
Since some programs exclude alignment columns with gaps, 
such programs will end up only with data from genes which are available for 
the whole set of species. Datasets with missing sequences are discarded. 

%%%%%%%%%%%%
\subsection{\emph{High Level} Methods}

While \emph{low level methods} combine the source datasets directly, 
\emph{high level methods} construct a tree for each source dataset.
The trees from the source datasets are then combined to an overall tree 
(Fig.~\ref{fig:combination-level}).
If all source datasets contain the full set of species, 
consensus techniques
can be applied combining all the equally sized trees into one consensus tree.
To combine sets of trees with unequal, but overlapping sets of species, 
so-called \emph{supertree} methods have been developed which amalgamate 
the input trees into one overall \emph{supertree}.

%%%%%%%%%%%%%%
\subsubsection{Combining Equal-Sized Datasets: Consensus Methods}
\label{sec:consensus}

\begin{figure}[t]
  \centering
  \includegraphics[width=\textwidth]{fig-missing/consensus-all}
  \caption[Consensus Methods]
          {Consensus methods}
  \label{fig:consensus-all}
\end{figure}

Consensus methods aim to construct a consensus tree from a set of 
trees preserving common topological details of the source trees.
Commonly applied consensus methods are strict consensus, 
majority-rule consensus \cite{margush1981}, semi-strict consensus 
\cite{bremer1990}, and Adams consensus \cite{adams1986}. 

Most consensus methods are based on splits or bipartitions 
(cf. \ref{sec:intro:treenota})
found in a majority of the source trees ($M_l$ consensus 
\emph{sensu} \cite{mcmorris1983}, where $l$ represents 
the percent occurrence of included splits). 
Sets of non-contradicting bipartitions are chosen to reconstruct 
the consensus tree. 

The \emph{majority-rule consensus} \cite{margush1981} uses all splits
occurring in the majority of input trees. Usually a \emph{majority-rule
consensus} is assumed ($M_l$ consensus with $l>0.5$, \cite{mcmorris1983}). 
This means all splits occurring in more than $50\%$ of the source trees 
(see Fig.~\ref{fig:consensus-all}). 
Considering $l>0.5$ ensures that all splits can be incorporated in a
tree topology, because no two contradicting splits are able to 
occur in more than $50\%$ of the input trees.

For the \emph{strict consensus} tree only bipartitions are chosen that
are found in all input trees ($M_{1.0}$, see Fig.~\ref{fig:consensus-all}).
The \emph{semi-strict consensus} \cite{bremer1990} contains all splits 
that are not contradicted in any of the input trees, e.g. the 
split $ABCD|EF$ is uncontradicted by the $ABC$-trifurcation in 
tree A and B in Fig.~\ref{fig:consensus-all}.
Note that such splits can occur in less than half of the source 
trees. If all trees are binary trees \emph{strict} and \emph{semi-strict
consensus} will always produce the same trees.

   \begin{figure}[t]
     \centering
     \includegraphics[width=.8\textwidth]{fig-missing/consensus-adams}
     \caption[Adams Consensus]
             {Adams Consensus}
     \label{fig:consensus-adams}
   \end{figure}

Contrary to most consensus methods which are based on bipartitions and their 
percent occurrence in the source trees, the \emph{Adams consensus} 
is based on common nestings in trees, i.e., taxa frequently occurring
together in the same subtree. This method tries to find 
groups of sequences that are commonly occurring together in the
source trees \cite{adams1986}. \emph{Adams consensus trees}
can contain groups that cannot be found in any of the source trees.
This makes it difficult to interpret. Yet it can be informative
when there are sequences that are difficult to place. Such sequences 
are moved to the root of the \emph{Adams consensus tree} 
(Fig.~\ref{fig:consensus-adams}).

%%%%%%%%%%%%%%%%%%%%%
\subsubsection{Combining Overlapping Datasets: Supertree Methods}

Supertree methods have been developed to combine sets of overlapping trees 
into \emph{'supertrees'} containing all leaves found in the source trees.
Some of the methods are restricted to rooted trees, based on the argument 
that the broad majority of trees published are rooted trees. 
Input trees cannot be produced from different datasets only, 
but can also be obtained from the literature.

%%%%%%%%%
\paragraph{Matrix Representation Methods}

\begin{figure}[t]
  \centering
  \includegraphics[width=.5\textwidth]{fig-missing/MRP-schemes}
  \caption[MRP Coding Schemes]
          {Coding schemes to encode tree topologies for MRP methods:
           Baum/Ragan and Purvis' scheme}
  \label{fig:MRP-schemes}
\end{figure}

A commonly used method to construct supertrees is 
\emph{Matrix Representation using Parsimony} 
(MRP, \cite{baum1992,ragan1992}), which uses binary coding to
represent the input trees. The splits induced by all source trees 
are coded into a matrix with binary characters ('0', '1') 
with missing data ('?').
The binary matrix is then used to construct an overall tree 
(cf. Fig.~\ref{fig:combination-level}) 
using \emph{Maximum Parsimony} methods \cite{swofford1996}. 
The most abundant coding schemes are those independently 
suggested by \cite{baum1992} and \cite{ragan1992} and the 
modified scheme by \cite{purvis1995}.
Baum/Ragan code the bipartitions 
of an input tree assigning '1' to all leaves in the one part of the 
bipartition and '0' to the other that contains the root. All missing taxa
are assigned the missing data character '?' (Fig. \ref{fig:MRP-schemes}).
\cite{purvis1995} differs in that only the sistergroup of a clade
is assigned '0', while all other leaves of that part outside the clade
are assigned '?' (Fig.~\ref{fig:MRP-schemes}).

To give input trees different weights in the analysis, different weighting
schemes have been suggested (see \cite{salamin2002}, 
\cite{sanderson1998}, and references herein for more details).

Recently, another method has been proposed called \emph{Matrix Representation
using Flipping} (MRF, \cite{chen2003}), which also uses a binary matrix 
coding of the trees as described above. The MRF supertree is constructed 
by 'flipping' conflicting character states from '1' to '0', or vice versa
to produce a matrix that does not contain contradictions.
Their optimal solution is the tree that needs the least 'flips'.


%%%%%%%%%
\paragraph{Direct Supertree Methods}

%\begin{figure}[p!]
%     \centering
%     \includegraphics[width=.8\textwidth]{fig-missing/mincut}
%     \caption[\textsc{MinCut Supertree} Algorithm]
%             {The \textsc{MinCut Supertree} Algorithm. 
%              For more details, see text.}
%     \label{fig:mincut}
%\end{figure}

%\begin{algorithm}[p!]
%\caption[\textsc{MinCut Supertree}]
%        {\textsc{MinCut Supertree}~(Set of rooted trees $\mcT$, 
%                                   set of $n$ leaves)}
%\label{alg:mincut}
%
%  \Switch{number of leaves $n$}{
%  \lCase{($n=1$)}{\KwRet a single node $x_1$\;}
%  \lCase{($n=2$)}{\KwRet a tree with 2 leaves $x_1$ and $x_2$\;}
%  \Other{
%\lnl{alg:mincut-ST}%
%      Construct the graph $S_{\mcT}$ by connecting
%      each pair of taxa $a$ and $b$ by an edge $e_{(a,b)}$ ~if~ they are 
%      connected in at least one tree in $\mcT$ by a path that does
%      not pass the root of that tree, i.e., $\MRCA(a, b) \neq root$
%      (cf. Fig.~\ref{fig:mincut}, $S_{\mathrm{\{Tree A,B\}}}$)\;
%
%      Weight the edge $e_{(a,b)}$ by the number of trees supporting 
%      $\MRCA(a, b) \ne root$\;
%  }}
%  \eIf{$S_{\mcT}$ is a disconnected graph}{
%      Let $S_i$ be the components of $S_{\mcT}$\;
%  }{
%      Construct $S_{\mcT}/E_{\mcT}^{Max}$ as follows: 
%      Merge all nodes $a$ and $b$ connected by an edge $e_{(a,b)}$ with 
%      maximum weight $|\mcT|$, i.e., supported by the whole set of trees
%      and remove edge $e_{(a,b)}$ (cf. Fig.~\ref{fig:mincut})\;
%
%\lnl{alg:mincut-cut}%
%      Remove all edges from $S_{\mcT}/E_{\mcT}^{Max}$ that are
%      a minimum cut set of $S_{\mcT}$.
%
%      Let $S_i$ be the resulting components of $S_{\mcT}/E_{\mcT}^{Max}$\;
%   }
%  \ForEach {component $S_i$}{ 
%     $T_i$ = ~\textsc{MinCut Supertree}~($\mcT|S_i, L=S_i$), 
%             where $\mcT|S_i$ denotes the trees from $\mcT$ with all leaves 
%             $r \not\in S_i$ pruned\;
%  }
%  Construct a new tree $T$ by connecting the roots of $T_i$ to a new tree\;
%  \KwRet $T$\
%\end{algorithm}


Another family of supertree methods use a more graph theoretical 
approach. Members of that family are 
  the \textsc{OneTree}/BUILD algorithm \cite{aho1981,bryant1995},
  \textsc{MinCut Supertree} algorithm \cite{semple2000}, and
  the modified \textsc{MinCut Supertree} algorithm
  (\textsc{ModMinCut Supertree}, \cite{page2002d}).
These algorithms take as input sets of rooted trees. The 
BUILD algorithm and the \textsc{OneTree} algorithm produce
a supertree only if the input trees are compatible, i.e., the trees
are not contradictory (cf. strict consensus in \ref{sec:consensus}).
Since this is almost never the case for biological data, these 
algorithms are mainly useful to test for compatibility of trees.

\textsc{MinCut Supertree} tries to yield a supertree applying 
a minimum cut algorithm to the graph $S_{\mcT}$ constructed
from the set of input trees $\mcT$, following 
Alg.~\ref{alg:mincut} (see also Fig.~\ref{fig:mincut} and
Fig.~\ref{fig:combination-level}). The notations follow \cite{page2002d} --
for further details see there.

%\begin{figure}[t]
%     \centering
%     \includegraphics[width=\textwidth]{fig-missing/modmincut}
%     \caption[\textsc{ModMinCut Supertree} Algorithm]
%             {The \textsc{ModMinCut Supertree} algorithm. 
%              For more details, see text.}
%     \label{fig:modmincut}
%\end{figure}

\cite{page2002d} suggested a modification (here called \textsc{ModMinCut}) 
to choose the minimum cuts, 
that takes into account preserving uncontradicted information from 
the source trees. The \textsc{ModMinCut Supertree} algorithm marks all edges 
in the graph $S_{\mcT}$ (step \ref{alg:mincut-ST}) not contradicted 
by any tree in $\mcT$. 
If $S_{\mcT}$ has to be cut in step \ref{alg:mincut-cut}
minimum cuts are preferred which do not cut uncontradicted edges.
Hence, the amount of uncontradicted information from the source trees
is preserved in the supertree (cf. Fig.~\ref{fig:modmincut}).

%%%%%%%%%%%%%
\paragraph{Quartet-Based Supertree}
Recently a quartet-based supertree method has been proposed by
\cite{robinson2001}. Instead of using a matrix representation, all source 
trees are decomposed into sets of quartets contained in the topologies
(Fig.~\ref{fig:combination-level}). 
Each quartet is weighted by the highest support value found on the 
path of the quartet's middle edge in any of the input trees (the authors used 
the TREE-PUZZLE package to compute the trees and support values).
From this set of weighted quartets an overall tree is constructed with
the quartet method implemented in the AllTree program \cite{bendor1998}.
In a first step, this method computes all subsets of taxa and keeps those 
satisfying the most quartets. In a second step, it performs an exact 
search on each subset constructing an overall tree satisfying most quartets.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{\emph{Medium Level} Combined Phylogenetic Analysis}
\label{sec:quartet-combin} 

%%%%%%%%%%%%
\subsection{Notation}
\label{sec:nota-incompl}

In the present chapter we assume a collection $\mcS = \{s_1, \ldots, s_n\}$ 
of $n$ species and $k$ different genes to be used for combined analysis.

With $\mcS_1, \mcS_2, \ldots,\mcS_k$ we denote subsets of $\mcS$, such that
\begin{eqnarray}
 \left| \mcS_g \right| \le 4 \mbox{\qquad for each \quad} g&
 \mbox{\qquad and \qquad}&
 \bigcup_{g=1}^k \mcS_g = \mcS.
\end{eqnarray}
Each $\mcS_g$ represents the subset of species for which a
multiple alignment based on gene $g$ is available. These subsets 
will be called genesets. Finally, $\mcQ_g , g=1, \ldots, k$ represent 
the corresponding sets of possible quartets.

Instead of reconstructing the trees $T(\mcS_g)$ for each set $\mcS_g$, 
we will compute an overall tree $T(\mcS)$ combining the
information provided from the evaluation of the quartet sets
$\mcQ_1, \mcQ_2, \ldots,\mcQ_k$ from the $k$ different alignments.
Note that a taxon is not represented by one sequence any more.

%%%%%%%%%%%%%%%%%
\subsection{The Combined Quartet Method to Combine Genesets}
\label{sec:missing:qp-combine}


%%%%%%%%%%%%%%%%%
\subsubsection{Combining the ML Quartets}

In the method proposed, the genesets will be combined on the level
of the quartets. As a guideline to combine the quartets we use 
the log-likelihood $\ell^{(g)}_{ab|cd}$, $\ell^{(g)}_{ac|bd}$, 
and $\ell^{(g)}_{ad|bc}$ for quartet $\{a,b,c,d\} \in \mcQ_g$
on the geneset $\mcS_g$.

To combine the datasets for each geneset $\mcS_1$, $\mcS_2$, 
\ldots $\mcS_k$, all three $\log$-likelihoods for each quartet in
$\mcQ_1$, $\mcQ_2$, \ldots $\mcQ_k$ are evaluated first 
(cf. \emph{ML step} in \ref{sec:qp-alg}).

Then, we compute
\begin{eqnarray}
  \label{eq:missing:tlh1}
  \ell_{ab|cd} & = & \sum_{g=1}^k \ell^{(g)}_{ab|cd} \\
  \label{eq:missing:tlh2}
  \ell_{ac|bd} & = & \sum_{g=1}^k \ell^{(g)}_{ac|bd}\\
  \label{eq:missing:tlh3}
  \ell_{ad|bc} & = & \sum_{g=1}^k \ell^{(g)}_{ad|bc}
\end{eqnarray}
for each quartet. For the sake of simplicity, the log-likelihoods
$\ell^{(g)}_{\tau}=0$ if the quartet with the sequences $\mcL(\tau)$
is not represented by alignment $g$.

Alternatively, one can also compute
\begin{eqnarray}
  \label{eq:missing:quartmean1}
  \overline{\ell_{ab|cd}} &=& \frac{\ell_{ab|cd}}
                           {|\left\{g:\{a,b,c,d\}\in\mcQ_g\right\}|}\\
  \label{eq:missing:quartmean2}
  \overline{\ell_{ac|bd}} &=& \frac{\ell_{ac|bd}}
                           {|\left\{g:\{a,b,c,d\}\in\mcQ_g\right\}|}\\
  \label{eq:missing:quartmean3}
  \overline{\ell_{ad|bc}} &=& \frac{\ell_{ad|bc}}
                           {|\left\{g:\{a,b,c,d\}\in\mcQ_g\right\}|}.
\end{eqnarray}
$\overline{\ell_{\tau}}$ can be viewed as the average support a 
quartet tree $\tau$ receives from the set of sequence alignments.
In case that a quartet is not represented by any alignment the averages
are set equal to zero. % and the quartet is ignored in the procedure.

%%%%%%%%%%%%%%%%%
\subsubsection{Computing the Overall Tree}

To reconstruct a phylogeny $T(\mcS)$ based on the collection of 
log-likelihoods from Equations \ref{eq:missing:tlh1} to \ref{eq:missing:tlh3} 
or \ref{eq:missing:quartmean1} to \ref{eq:missing:quartmean3} we apply
the PUZZLE idea from \ref{sec:qp-alg}.
The straightforward application of the \emph{QP} algorithm is only possible 
if information exist for each quartet in $\mcQ$. This, for instance, 
is the case if at least one alignment $\mcS_g$ comprises all the species 
in $\mcS$. 

If some quartets are missing, which usually happens if each $\mcS_g$ is 
a proper subset of $\mcS$, then these quartets are treated as unresolved. 
A quartet tree is selected randomly among the three possible topologies.
In this case it is necessary to examine whether the overlapping genesets 
can be combined. This will be explained in the following.


%%%%%%%%%%%%%%%%%
\subsubsection{Assessing Whether Genesets Can Be Combined}


To combine two genesets $\mcS_i$ and $\mcS_j$ 
a minimum pairwise overlap of 3 sequence among the two subsets is required, 
\begin{equation}
   \left| \mcS_i \cap \mcS_j \right| \ge 3
   \label{eq:missing:pairoverlap} \mbox{.}
\end{equation}
We call Eq.~\ref{eq:missing:pairoverlap} the \emph{pair-overlap condition}.
The \emph{pair-overlap condition} is different from the 
\emph{overlap condition} (Eq.~\ref{eq:overlap}) in section \ref{sec:modifiedqp} 
where the subsets were constructed from one complete set of sequences 
to ensure combinability of the subsets.
Here the subsets are genesets. Their sizes and overlap are defined by 
the availability of sequence data for the species in $\mcS$. 
Note, that not every pair of genesets might share a sufficient overlap.

To assess whether these gene datasets can be combined, we construct an overlap
graph $G_{\mathit{ovl}}=(\mcV,\mcE)$ with a set of nodes $\mcV$ (genesets) and a set of 
undirected edges $\mcE$ (overlaps) with
\begin{eqnarray}
  \mcV & = & \{\mcS_1, \mcS_2, \ldots \mcS_k\}\\
  \mcE & = & \{e_{(\mcS_i,\mcS_j)}\mbox{ for }\left|\{\mcS_i \cap \mcS_j\}\right| \ge 3\}
  \quad 1\le i < j \le k.
\end{eqnarray}
The edge weights consist of the amount of overlap fulfilling the 
\emph{pairwise-overlap condition} (see for an example 
Fig.~\ref{fig:missing:overlap}). This representation of the 
datasets provides insight into several properties of the entire dataset. 

If the graph consists of unconnected subgraphs, not all data is suited 
for a combined analysis. According to the \emph{pairwise-overlap condition}
gene datasets from different connected components cannot be used 
simultaneously in the same analysis because no quartet information is 
are available to reasonably guide the insertion of the sequences.

Genesets from one connected component, however, can be used for combined 
analysis. Although genesets might exist within a connected component which
do not share sufficient overlap with each other to be combined, they can be linked
by means of other genesets which first have to be added 
satisfying the \emph{pairwise overlap condition}, to connect the two.

Note, that by applying the \emph{overlap condition} from Eq.~\ref{eq:overlap},
sets from one connected component can gain an overlap $\ge 3$
to the leafset $\mcL(T_i)$ of a tree $T_i$ so far reconstructed
from genesets of another connected component. 
Nevertheless, combining those sets will produce a very doubtful tree,
because too little information is available to guide the
insertion of sequences.

The combinability of the genesets from a connected component is characterized
by two features of the \emph{overlap graph}. A high connectivity
within a component (each geneset shows overlap to many other geneset)
results in network-like quartet information among the genesets.
This reduces the need for mediating sets to connect non-overlapping genesets.
Even more important is the size of the overlap. The higher the overlap,
the more quartets are shared among two neighboring genesets. 
Consequently, a higher amount of information is available to guide the
insertion of sequences from a geneset into a tree already reconstructed. 

%%%%%%%%%%%%%%%%%
\subsubsection{Overlap-Guided Puzzling Step}

As described above, sufficient overlapping information is needed 
to reasonably insert a new leaf $s_{i+1}$ into a tree $T_i$. Hence,
instead of using a random permutation of leaves, the permutation
has to follow certain restrictions. 
To satisfy the \emph{pairwise overlap condition}, 
we apply a graph based procedure similar to Prim's minimum spanning tree 
(MST) algorithm (see, e.g., \cite{cormen2001}) assuming equal edge weights.
We start with the leaves from a randomly picked geneset 
$\mcS' \in \{\mcS_1\ldots \mcS_k\}$. 
We define a front set $\mcF$ containing all unused nodes (genesets) 
from the overlap graph that are connected by an edge to any geneset 
already used in the tree. 
From the sequences in $\mcS'$ we construct the tree $T(\mcS')$
applying the usual puzzling step algorithm 
(see section \ref{sec:qp-alg} and chapter \ref{chap:complex}).
Then we randomly draw one geneset $\mcS''$ from $\mcF$. We remove 
$\mcS''$ from $\mcF$ and add the sets connected
to $\mcS''$, but not yet used to $\mcF$. The sequences of $\mcS''$ 
are then added to the tree. 
Guided by the overlap graph we thus construct an 
intermediate tree by sequentially adding the genesets. 
As in the usual \emph{puzzling step} many intermediate trees are 
constructed using different orders of leaves and genesets.

%%%%%%%%%%%%
\subsubsection{Relative Majority Consensus}
\label{sec:missing:mrel}

Since missing data adds more ambiguity, this naturally hampers the 
construction of a resolved majority-rule consensus tree. We therefore 
decided to apply a majority consensus which is slightly different 
from the majority-rule consensus used in section \ref{sec:qp-alg}. 
Instead of using only splits occurring in more than $50\%$ 
($M_l$ with $l>0.5$) of the intermediate trees, also splits 
below $50\%$ are considered. 
The aim is to use as many splits as are supported 
by relative majority to extract a maximum 
uncontradicted information from the intermediate trees.

The relative majority consensus $M_{\mathit{rel}}$ adds all congruent
splits from the set of intermediate trees in descending order of occurrence.
No splits will be accepted for the consensus that have the same or
lower percentage occurrence as any incongruent split.
This consensus procedure does not imply a fixed threshold but 
uses a variable percentage down to the first incongruence 
guided by the relative majority. 

